Author's response to reviews

Title: Reshaping maternal services in Nigeria: any need for spiritual care?

Authors:

Abiodun Idowu Adanikin (adanikinbiodun@yahoo.com)
Uche Onwudiegwu (uonwudie@yahoo.com)
Akinyemi Akinsoji Akintayo (akinyemiakintayo@yahoo.co.uk)

Version: 5 Date: 24 May 2014

Author's response to reviews: see over
The Editor,
BMC Pregnancy and Childbirth,

Dear Sir/Ma,

**COVERING LETTER**

On behalf of the authors of the article ‘Reshaping maternal services in Nigeria: any need for spiritual care?’ I write to submit a revised version of the manuscript. The point-by-point corrections as requested by the editorial board have been addressed as follows:

1. Formatting changes:
   a. Title page: email address of all authors have been included
   b. List of abbreviations: has been appropriately added

2. Editorial comment: Reference for the practical tool used to develop the questionnaire and its validation have been discussed in paragraph 5 of the Methodology.

Thank you.

Yours faithfully,

Adanikin Abiodun I.